Author's response to reviews

Title: Clear lens phacoemulsification in anterior lenticous due to Alport syndrome: a case series

Authors:

   Ghassem Amiraslanzadeh (dr.aslanzadeh@gmail.com)
   Davood Gharabaghi (gharabaghid@gmail.com)
   Niloofar Naderi (dr_niloofarnaderi@yahoo.com)

Version: 10 Date: 21 December 2007

Author's response to reviews: see over
Cover letter 21Dec.
I tried to edit the whole manuscript and include valuable comments of reviewers, Dr Radcliffe and Dr Saxena:
Corrections or removing of informal sentences, consistent use of scientific abbreviations, and revision of grammar and quality of English language. These are the point-by-point corrections and alterations done:
Dr Radcliffe’s comments:
- Purpose > to describe...
- VA = visual acuity
- IOL = intraocular lens...all edited
- General anesthesia...hearing problem...corrected
- Fortunately> There were no...Corrected
- The same procedure...
- Typo...corrected
- Quality of English...revised by somebody else.
Dr. Saxena's comments:
- Abbreviations must be consistent...done
- Case 2 p 2 line 3 Add...
- Page 3 line 4 dual sensory loss...(seeing and hearing loss)
- Page 3 line 12 ...revised
- Page 3 line 15 ...added
- Page 3 line 23...changed
- Page 3 conclusion...deleted and changed,
- References...changed
- Quality of English revised.